Author's response to reviews

Title: The efficacy and security of Jianweiqutong pill on the treatment of chronic nonatrophic gastritis (spleen and stomach qi deficiency with damp-heat stasis syndrome): study protocol for a phase II, randomized controlled trial.

Authors:

Xiaoxin Zhang (373020277@qq.com)
Weiwei Chen (cww1984@126.com)
Bin She (She-bin@163.com)
Ruijie Luo (312167554@qq.com)
Na Shi (785239482@qq.com)
Ping Xue (duburongcheng@163.com)
Xiaonan Yang (yangxxnan@163.com)
Qing Xia (xiaqing@medmail.com.cn)

Version: 6 Date: 18 June 2014

Author's response to reviews:

1. The statement of ‘Jianweiqutong micro-pills’ were corrected as ‘Jianweiqutong pill’.
2. The statement of ‘qi deficiency of spleen and stomach together with dampness-heat and blood stasis syndrome’ (QDSSDBS) were corrected as ‘spleen and stomach Qi deficiency with damp-heat stasis syndrome#SWQDDSS#’.
3. The statement of ‘Fangsheng Pharmaceutical Co Ltd, Hunan, China’ were corrected as ‘Hunan Anbang pharmaceutical Limited by Share Ltd, China’.
4. We have added the reference numbers given with ethical approval in the Methods section.

We tried our best to improve the manuscript and made some other changes in the manuscript. These changes will not influence the content and framework of the paper. And here we did not list the changes but marked in red in revised paper.